Author's response to reviews

Title: Effectiveness of First-Line Antiretroviral Therapy and Correlates of Longitudinal Changes in CD4 and Viral Load Among HIV-infected Children in Ghana

Authors:

Oliver Barry (omb2104@columbia.edu)
Jonathan Powell (jlp2005@med.cornell.edu)
Lorna Renner (lornarenner@gmail.com)
Evelyn Bonney (EBonney@noguchi.mimcom.org)
Meghan Prin (Meghan.prin@gmail.com)
William Ampofo (wampofo@noguchi.mimcom.org)
Jonas Kusah (jonaskusa@yahoo.co.uk)
Bamenla Goka (bamenla@yahoo.co.uk)
Kwamena Sagoe (kwamenawcs@yahoo.com)
Veronika Northrup (veronika.northrup@yale.edu)
Elijah Paintsil (elijah.paintsil@yale.edu)

Version: 2 Date: 5 September 2013

Author's response to reviews: see over
September 5, 2013

Dear Editor:

Thank you for sending out our manuscript for review. We were highly encouraged by the overall favorable response the manuscript received. The manuscript has been modified according to the comments and suggestions made by the reviewers. See a rebuttal of all the issues raised below.

Thank you for considering our paper for publication in BMC Infectious Diseases Journal.

Sincerely,

[Signature]

Elijah Paintsil, MD
Departments of Pediatrics and Pharmacology
Yale School of Medicine
464 Congress Ave
New Haven, CT 06520
Phone: (203) 785-6101, Fax: (203) 785-6961
Email: elijah.paintsil@yale.edu